Projected Global Temperature Changes after Net Zero are Small but Significant

Alexander R Borowiak¹, Andrew David King¹, Josephine R. Brown¹, Christopher David Jones², Tilo Ziehn³, Malte Alexander Meinshausen¹, and Liam Cassidy⁴

¹University of Melbourne ²Met Office Hadley Centre ³Commonwealth Scientific and Industrial Research Organisation (CSIRO) ⁴The University of Melbourne

February 27, 2024

Abstract

As more countries make net zero greenhouse gas emissions pledges, it is crucial to understand the effects on global climate after achieving net zero emissions. The climate has been found to continue to evolve even after the abrupt cessation of CO2 emissions, with some models simulating a small warming and others simulating a small cooling. In this study, we analyse if the temperature and precipitation changes post abrupt cessation of CO2 emissions are significant compared to natural climate variations. We find that the temperature changes are outside of natural variability for most models, whilst the precipitation changes are mostly non-significant. We also demonstrate that post-net zero temperature changes have implications for the remaining carbon budget. The possibility of further global warming post-net zero adds to the evidence supporting more rapid emissions reductions in the near-term.

Hosted file

zec_natural_var_grl.docx available at https://authorea.com/users/537888/articles/717362projected-global-temperature-changes-after-net-zero-are-small-but-significant

Hosted file

supplementary_material_grl.docx available at https://authorea.com/users/537888/articles/ 717362-projected-global-temperature-changes-after-net-zero-are-small-but-significant









Figure 1.



1	Projected Global Temperature Changes after Net Zero are Small but Significant
2	
3	Alex Borowiak ^{1, 2} , Andrew King ^{1, 2} , Josephine Brown ^{1, 2} , Chris Jones ³ , Tilo Ziehn ⁴ , Malte
4	Meinshausen ¹ , Liam Cassidy ¹
5	
6	1. School of Geography, Earth and Atmospheric Sciences, University of Melbourne,
7	Parkville, Victoria, Australia.
8	2. Australian Research Council Centre of Excellence for Climate Extremes
9	3. Met Office Hadley Centre, Exeter EX1 3PB, United Kingdom
10	4. CSIRO Environment, Aspendale, Victoria, Australia
11	Corresponding author: Alex Borowiak (aborowiak@student.unimelb.edu.au)
12	Key Points
13	• Changes in global mean surface temperature after the abrupt cessation of emissions are
14	significant compared to natural variability.
15	• Global mean precipitation changes after 50 years are only significant compared to natural
16	variability in models that warm.
17	• The temperature changes post-net zero have uncertain implications for the remaining
18	carbon budget.
19	

20 Abstract

21 As more countries make net zero greenhouse gas emissions pledges, it is crucial to understand 22 the effects on global climate after achieving net zero emissions. The climate has been found to 23 continue to evolve even after the abrupt cessation of CO₂ emissions, with some models 24 simulating a small warming and others simulating a small cooling. In this study, we analyse if the 25 temperature and precipitation changes post abrupt cessation of CO₂ emissions are significant 26 compared to natural climate variations. We find that the temperature changes are outside of 27 natural variability for most models, whilst the precipitation changes are mostly non-significant. 28 We also demonstrate that post-net zero temperature changes have implications for the remaining 29 carbon budget. The possibility of further global warming post-net zero adds to the evidence 30 supporting more rapid emissions reductions in the near-term.

31

32 Plain Language Summary

33

34 As more countries commit to achieving net-zero greenhouse gas emissions, it is essential to 35 understand the impact this will have on the global climate beyond this point. It has been found 36 that even after CO₂ emissions are abruptly halted, the climate continues to change, with various 37 models predicting either a slight warming or cooling effect. In our study, we investigate whether 38 the temperature and precipitation changes that occur after the sudden cessation of CO₂ 39 emissions are noteworthy when compared to natural climate variations. Our analysis reveals that 40 the temperature changes, in the majority of models, surpass what can be attributed to natural 41 variability. However, the precipitation changes are generally not significant. Additionally, we

42 observe that the temperature changes post-net zero have implications for the remaining carbon43 budget.

44

45 Introduction

46

Global greenhouse gas emissions must be reduced to near zero to prevent continued global
warming (MacDougall et al., 2020; Matthews & Zickfeld, 2012). This is required if the target of
"[...] holding the increase in the global average temperature to well below 2°C above
preindustrial levels and pursuing efforts to limit the temperature increase to 1.5°C [...]"
(UNFCCC, 2015) outlined by the Paris Agreement is to be achieved. As such, many countries
have made net zero pledges (*Energy & Climate Intelligence Unit* | *Net Zero Scorecard*, n.d.).

53

54 Several experiments have been run that aim to quantify the response of the global and local 55 climate after the cessation of emissions (Dvorak et al., 2022; Jones et al., 2019; Sherwood et al., 56 2022). One such experiment, the Zero Emissions Commitment Model Intercomparison Project 57 (ZECMIP)(Jones et al., 2019), aims to understand the evolution of the climate after the abrupt cessation of CO₂ emissions at around 1.5°C. In this scenario, Earth System Models (ESMSs) and 58 59 Earth System Models of Intermediate Complexity (EMICs) predict a post-net zero global 60 average temperature change of -0.07°C (-0.36°C to 0.29°C between models) 50 years after 61 emission cease - this is the Zero Emissions Commitment (ZEC) (MacDougall et al., 2020). A 62 warming may occur due to the thermal inertia of the oceans, causing further increases in the 63 global average temperature. Cooling may occur due to carbon dioxide removal by the terrestrial 64 biosphere and ocean. Ultimately, the trajectory of the climate after the cessation of emissions,

and whether the climate will warm or cool, depends on the magnitude of these two effects(MacDougall et al., 2020, 2022).

67

68 As the ZEC is small, it is important to know if this is robustly distinct from the background 69 climate variations. Previous studies have averaged ZEC across a multi-model ensemble but not 70 assessed the internal variability of each model which is known to cause apparent trends on short 71 timescales. Here, we analyse whether global temperature changes after the abrupt cessation of 72 CO₂ emissions are significant for several ESMs compared to the natural variability of their 73 climate. We also analyse if the global mean precipitation changes post-net zero are significant 74 compared to their natural variability. 75 76 Any post-net zero changes will have implications on the remaining carbon budget (RCB). The 77 RCB is the cumulative CO₂ that can be emitted while keeping the peak global average 78 temperature rise below a global warming level (Dvorak et al., 2022). The RCB allows emissions 79 reduction targets to be aligned with global warming levels. If annual emissions remain at 2022 80 levels (40.2 GtCO₂), nine years (380 GtCO₂) remain in the carbon budget from the beginning of 81 2023 for a 50% chance of avoiding exceeding 1.5°C (Global Carbon Project (GCP), n.d.). In 82 previous work that used the ZEC when quantifying the RCB, the ZEC value is either ignored (Matthews et al., 2020), considered zero¹¹ or has an uncertainty distributed around zero 83 84 (Matthews et al., 2021; IPCC, 2021). Here we calculate the impact of the ZEC on the RCB. 85

86 Model Data

87

ZECMIP Simulations

90	To analyse the temperature changes post-net zero, we used the Zero Emission Commitment
91	Model Intercomparison Project (ZECMIP) A1 experiment (Jones et al., 2019) - the highest-
92	priority experiment with more models available. Emissions in this experiment increase by 1% per
93	year until 1000 PgC has been released, at which point CO2 emissions are abruptly ceased. Nine
94	Earth System Models have run the ZECMIP A1 experiment: NorESM2-LM (Tjiputra et al.,
95	2020), MIROC-ES2L (Hajima et al., 2020), MPI-ESM1-2-LR (Mauritsen et al., 2019), GISS-E2-
96	1-G-CC (Kelley et al., 2020), GFDL-ESM4 (Dunne et al., 2020), ACCESS-ESM1-5 (Law et al.,
97	2017; Ziehn et al., 2020), CESM2 (Danabasoglu et al., 2020; Lawrence et al., 2019), UKESM1-0-
98	LL (Sellar et al., 2019) and CanESM5 (Swart et al., 2019). All models are run as part of the
99	Community Model Intercomparison Project Phase 6, and each contains an interactive carbon
100	cycle. For a full summary of model features, see MacDougall et al., (2020). The global mean
101	temperature anomalies and global mean precipitation anomalies after the cessation of emissions
102	for each model can be seen in Supplementary figures 1 and 2. We do not include EMICs in this
103	analysis as they do not have a realistic representation of interannual climate variability.
104	Precipitation is not available for GFDL and NorESM2 models.
105	
106	Pre-Industrial Control Simulations
107	
108	Pre-industrial control simulations are initialised with greenhouse gas levels from the reference
109	year 1850 ⁷ . This year is selected as it precedes the commencement of large-scale industrialisation.

Pre-industrial control simulations illustrate the climate's natural variability without human

111 interference. The global mean temperature and global mean precipitation for all pre-industrial

112 control simulations can be seen in Supplementary figures 3 and 4.

113

114 Methods

115

116 **ZEC Calculation**

117

118 The temperature and precipitation changes after the cessation of emissions are compared with 119 the twenty-year average of the point at which the A1 experiment branches from the 1% CO₂ run. 120 The ZEC₂₅ and ZEC₅₀ values are then calculated as the 20-year average centred on the years 25 121 and 50, respectively. To compare these values with the range in the pre-industrial control, we 122 calculate the difference between two twenty-year average periods separated by five years corresponding to ZEC₂₅, and 30 years corresponding to ZEC₅₀. These values can then be used to 123 124 create a distribution, as shown in figure 1. Values are considered outside the range of natural variability if they are below the 1st percentile or above the 99th percentile of the pre-industrial 125 126 control anomalies. 127

128 Carbon Budget Calculation

129

130 The carbon budget is defined as the allowable remaining emissions for keeping the peak global

131 average temperature below a certain global warming level. The allowable emission can be derived

132 using a known linear relationship between warming and cumulative CO₂ emissions (Seshadri,

133 2017). This relationship is commonly estimated using the transient response to cumulative

emissions (TCRE). TCRE change in the global mean temperature (ΔT) to increasing cumulative emissions (E).

$$TCRE = \frac{\Delta T}{E}$$

136

In this study, we have used a common method of estimating this parameter from the 1% CO_2 simulations by taking the 20-year average once 1000 PgC has been emitted. The 20-year average can then be converted to an anomaly by subtracting the average of the pre-industrial control simulation (all years for each model). This gives the sensitivity of the climate to increasing cumulative emissions (°C/GtCO₂), which can then be used to calculate the remaining emissions for a given peak global warming. In this study, we use this relationship to infer the associated emissions (*Budget_c*) with a given ZEC₅₀

$$Budget_{C} [GtCO_{2}] = ZEC_{50} [°C] * \frac{3670}{TCRE [°C/1000PgC]}$$

144 **Results**

145

146 The temperature changes after cessation of emissions are outside the range of natural variability 147 in most models (figure 1 left column) (see Supplementary Table 1 for a summary of how the 148 ZEC values compare with natural variability). Only ACCESS-ESM1-5 and two of the CanESM5 149 ensemble members show non-significant changes, with these models predicting little global 150 average temperature change. Only two models simulate warming (ACCESS-ESM1-5 and 151 UKESM1-0-LL); however, this warming is only significant compared to natural variability in the 152 UKESM1-0-LL model. The remainder of the models cool, and the cooling, both after 25 and 50 153 years, is simulated to be lower than the range due to natural variability (except for GISS-E2-1-G-154 CC at 25 years, GFDL-ESM4 after 50 years and two of the ensemble members of CanESM5).

- 155 Thus, the response most commonly simulated after the cessation of emissions is a cooling that is
- 156 significant compared to natural variability, based on the model simulations examined here.
- 157







- 160 separated by 5 years (blue) and 30 years (purple) for each model (rows), compared with the ZEC₂₅ (blue line(s)) and ZEC₅₀ (purple lines(s)).
- 161 *Precipitation is not available for GFDL and NorESM2 models.*

We next analyse the post-abrupt CO_2 emissions cessations effect on precipitation (figure 1 right column). The simulations show significant increases in precipitation in all models after 25 years, except MPI-ESM1-2-LR. By the year 50, precipitation changes are still positive for all models,

165 excluding MPI-ESM1-2-LR; however, these changes are not significant for most models. As 166 atmospheric CO₂ concentrations decline, the reverse of what is described in Andrews et al., 167 (2009) may occur. The decreasing CO_2 will decrease the positive radiative component at the top 168 of the atmosphere greater than the surface, increasing latent heat flux, resulting in increased 169 evaporation, and more rainfall. This effect may counteract or add to the Clausius Clapeyron 170 effect, which results in air temperature increases as surface temperatures increase. The only two 171 models that see significant increases in precipitation after 50 years (ACCESS-ESM1-5 and UKESM1-0-LL), are both models that see increased global mean surface temperature (although 172 173 ACCESS-ESM1-5 warming is non-significant). The decreased CO_2 and increased surface 174 temperature, thus both act to increase precipitation. For all other models, the precipitation 175 change is significant at year 25 compared to natural variability likely due to the inertia of global 176 average precipitation that persists even after the global temperature is no longer increasing 177 (Mitchell et al., 2016). However, by the year 50, as CO₂ concentrations are still decreasing in 178 models, and surface temperature has decreased, these effects are likely to have counteract, 179 resulting in non-significant precipitation changes. Additionally, the variability of precipitation is larger compared to temperature, and thus, the forced changes in precipitation must be greater in 180 181 order to be significant relative to natural variability (Milinski et al., 2020). Our findings show that 182 whilst the temperature changes after zero CO₂ emissions have a discernible influence on short-183 term precipitation patterns, longer-term changes in precipitation do not exceed the bounds of 184 natural variability.

185 These post-net zero temperature changes may have implications for the RCB. The multi-model

186 average RCB for 1.5°C, based upon current warming of 1.2°C (Globalwarmingindex.Org -- Tracking

187 *Progress to a Safe Climate*, n.d.) is 535 GtCO₂ (see Supplementary Table 2 for details on each

188 model). This value is larger than other reported values (Friedlingstein et al., 2022; Global Carbon

189 Project (GCP), n.d.; IPCC, 2021; Matthews et al., 2021) however, is still within the range of

190 possible values considering the large spread in the RCB and the limited number of models

191 available for this analysis. Should the climate exhibit a significant warming post-cessation of 192 emissions, a scenario identified here solely in the UKESM1-0-LL model, this results in a 193 reduction of the carbon budget by 518 GtCO₂. Consequently, with the RCB estimate being 525 194 GtCO₂ from this subset of models, this suggests that even if CO₂ emissions ceased, the climate 195 could still approach the 1.5°C warming level. In models that predict a significant cooling after 196 zero emissions, an average temperature reduction of 0.21°C is simulated. This cooling could 197 potentially increase the RCB for limiting global warming to 1.5°C by an additional 475 GtCO₂, representing a substantial 75% boost to the available carbon budget. However, this extra carbon 198 199 budget is only applicable under certain conditions. If CO2 emissions were to be abruptly halted, 200 the cooling effect would have no impact on the RCB. This is because the RCB is for the peak 201 global mean temperature, and a negative ZEC would not reduce the peak temperature reached 202 (see Supplementary figure 1). Rather, the negative ZEC will influence the climate evolution after 203 the global mean temperature has peaked, returning the global mean temperature to stabilisation 204 at a lower level. In reality, the reduction in emissions is likely to be a gradual and phased process. 205 Some of the ZEC may be realised before reaching net-zero emissions (Koven et al., 2023), 206 impacting the RCB. It is important to emphasise that the 475 GtCO₂ represents a maximum 207 potential increase for the RCB for 1.5°C, and the actual impact of ZEC on the RCB can vary 208 depending on the emissions reduction trajectory and the cumulative emissions (Allen et al., 209 2022). There is currently a lack of available simulations to precisely determine how much of the 210 ZEC effect will be realised with different pathways to net-zero emissions.

211 Discussion

212 Previous modelling studies have found that the changes in global average temperature after the

213 immediate cessation of CO_2 emissions are small (global average range of -0.36°C to

214 0.29°C)(MacDougall et al., 2020), but it has not been investigated if this is only due to internal

215 variability. The IPCC assessed that the changes after zero emissions are '[...] small compared

216 with natural variability in GSAT [Global Surface Air Temperature]' (IPCC, 2021) but this was 217 after averaging across the ensemble before making that comparison. Here, by comparing each 218 model with its own natural variability we find that for most models in the ZECMIP A1 219 experimental ensemble, the changes in global average temperature are significant. The response 220 is varied between models; however, the most commonly simulated response is a significant 221 cooling compared to natural variability - seven out of nine models (only one of three simulations 222 from CanESM5) cool significantly 50 years after the cessation of CO₂ emissions. However, the 223 cooling after emissions are abruptly ceased does not affect the RCB for 1.5°C, but may return 224 global average temperatures to a lower global warming level, reducing impacts from climate 225 change. Warming is less likely after the cessation of emissions, with only one model (UKESM1-226 0-LL) simulating significant warming compared to natural variability. This low-likelihood but 227 high-impact outcome means we must still plan ambitious emissions reductions to avoid the 228 possibility of exceeding the Paris Agreement warming levels following emissions cessation. Our 229 study underlines the need for improved understanding and constraints on TCRE and ZEC, and 230 we recommend larger ensembles of simulations are needed to enable more robust quantification 231 of the magnitude of ZEC in the face of climate variability, as well as less idealised experiments to 232 explore implications of net-zero.

233

The temperature and precipitation changes after net zero across climate models were found to be diverse, but the small ensemble of model simulations prevents robust probabilistic analysis of different outcomes under net-zero simulations. Given humanity's goal of reaching net zero emissions to prevent further global warming, it is imperative that the climate science community makes rapid advances in understanding the committed climate changes following net zero. Decision-makers need more information about the response of the climate to net zero emissions to plan accordingly. In the meantime, it may be prudent to account for the possibility of post-net zero emissions global warming and to take more rapid action to reduce greenhouse gas emissionsas a result.

243 Acknowledgments

- ARB was supported by the Australian Research Council Grant CE170100023. ADK, JRB, TZ
- 245 and MM were supported by the Australian Government's National Environment Science
- 246 Program. The research was completed with the assistance of resources from the National
- 247 Computational Infrastructure (NCI), which is supported by the Australian Government. Open
- 248 access publishing facilitated by The University of Melbourne, as part of the Wiley The
- 249 University of Melbourne agreement via the Council of Australian University Librarians.

250 Open Research

- 251 Data for the ZECMIP A1 experiment can be found at https://esgf.nci.org.au/projects/esgf-
- 252 <u>nci/</u>. The final code used to conduct this analysis will be published after any revisions that need
- to be made.

254			
255			
256			

258	
259	References
260	
261	
262	Allen, M. R., Friedlingstein, P., Girardin, C. A. J., Jenkins, S., Malhi, Y., Mitchell-Larson, E.,
263	Peters, G. P., & Rajamani, L. (2022). Annual Review of Environment and Resources Net Zero:
264	Science, Origins, and Implications. https://doi.org/10.1146/annurev-environ-112320
265	Andrews, T., Forster, P. M., & Gregory, J. M. (2009). A surface energy perspective on climate
266	change. Journal of Climate, 22(10), 2557–2570. https://doi.org/10.1175/2008JCLI2759.1
267	Danabasoglu, G., Lamarque, J. F., Bacmeister, J., Bailey, D. A., DuVivier, A. K., Edwards, J.,
268	Emmons, L. K., Fasullo, J., Garcia, R., Gettelman, A., Hannay, C., Holland, M. M., Large,
269	W. G., Lauritzen, P. H., Lawrence, D. M., Lenaerts, J. T. M., Lindsay, K., Lipscomb, W. H.,
270	Mills, M. J., Strand, W. G. (2020). The Community Earth System Model Version 2
271	(CESM2). Journal of Advances in Modeling Earth Systems, 12(2), e2019MS001916.
272	https://doi.org/10.1029/2019MS001916
273	Dunne, J. P., Horowitz, L. W., Adcroft, A. J., Ginoux, P., Held, I. M., John, J. G., Krasting, J. P.,
274	Malyshev, S., Naik, V., Paulot, F., Shevliakova, E., Stock, C. A., Zadeh, N., Balaji, V.,
275	Blanton, C., Dunne, K. A., Dupuis, C., Durachta, J., Dussin, R., Zhao, M. (2020). The
276	GFDL Earth System Model Version 4.1 (GFDL-ESM 4.1): Overall Coupled Model
277	Description and Simulation Characteristics. Journal of Advances in Modeling Earth Systems,
278	12(11), e2019MS002015. https://doi.org/10.1029/2019MS002015
279	Dvorak, M. T., Armour, K. C., Frierson, D. M. W., Proistosescu, C., Baker, M. B., & Smith, C. J.
280	(2022). Estimating the timing of geophysical commitment to 1.5 and 2.0 °C of global

- 281 warming. Nature Climate Change, 12(6), 547–552. https://doi.org/10.1038/s41558-022-
- 282 01372-у
- 283 Energy & Climate Intelligence Unit | Net Zero Scorecard. (n.d.). Retrieved August 14, 2023, from
- 284 https://eciu.net/netzerotracker
- 285 Friedlingstein, P., O'sullivan, M., Jones, M. W., Andrew, R. M., Gregor, L., Hauck, J., Le Quéré,
- 286 C., Luijkx, I. T., Olsen, A., Peters, G. P., Peters, W., Pongratz, J., Schwingshackl, C., Sitch,
- 287 S., Canadell, J. G., Ciais, P., Jackson, R. B., Alin, S. R., Alkama, R., ... Zheng, B. (2022).
- 288 Global Carbon Budget 2022. Earth System Science Data, 14(11), 4811–4900.
- 289 https://doi.org/10.5194/essd-14-4811-2022
- 290 Global Carbon Project (GCP). (n.d.).
- 291 globalwarmingindex.org -- Tracking progress to a safe climate. (n.d.). Retrieved October 11, 2023, from
 292 https://www.globalwarmingindex.org/
- 293 Hajima, T., Watanabe, M., Yamamoto, A., Tatebe, H., Noguchi, A., Abe, M., Ohgaito, R., Ito,
- A., Yamazaki, D., Okajima, H., Ito, A., Takata, K., Ogochi, K., Watanabe, S., & Kawamiya,
- 295 M. (2020). Description of the MIROC-ES2L Earth system model and evaluation of its
- 296 climate-biogeochemical processes and feedbacks. *Geosci. Model Dev. Discuss.*
- 297 https://doi.org/10.5194/gmd-2019-275
- 298 IPCC. (2021). Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the
- 299 Sixth Assessment Report of the Intergovernmental Panel on Climate Change: Vol. In Press. Cambridge
- 300 University Press. https://doi.org/10.1017/9781009157896
- 301 Jones, C., Frölicher, T., Koven, C., Macdougall, A., Matthews, D., Zickfeld, K., Rogelj, J., &
- 302 Tokarska, K. (2019). ZEC-MIP: Quantifying the Zero Emissions Commitment.

303	Jones, Chris., Frölicher, T. L., Koven, C., MacDougall, A. H., Damon Matthews, H., Zickfeld,
304	K., Rogelj, J., Tokarska, K. B., Gillett, N. P., Ilyina, T., Meinshausen, M., Mengis, N.,
305	Séférian, R., Eby, M., & Burger, F. A. (2019). The Zero Emissions Commitment Model
306	Intercomparison Project (ZECMIP) contribution to C4MIP: Quantifying committed
307	climate changes following zero carbon emissions. Geoscientific Model Development, 12(10),
308	4375-4385. https://doi.org/10.5194/gmd-12-4375-2019
309	Kelley, M., Schmidt, G. A., Nazarenko, L. S., Bauer, S. E., Ruedy, R., Russell, G. L., Ackerman,
310	A. S., Aleinov, I., Bauer, M., Bleck, R., Canuto, V., Cesana, G., Cheng, Y., Clune, T. L.,
311	Cook, B. I., Cruz, C. A., Del Genio, A. D., Elsaesser, G. S., Faluvegi, G., Yao, M. S.
312	(2020). GISS-E2.1: Configurations and Climatology. Journal of Advances in Modeling Earth
313	Systems, 12(8), e2019MS002025. https://doi.org/10.1029/2019MS002025
314	Koven, C. D., Sanderson, B. M., & Swann, A. L. S. (2023). Much of zero emissions commitment

315 occurs before reaching net zero emissions. *Environmental Research Letters*, 18(1).

316 https://doi.org/10.1088/1748-9326/acab1a

317 Law, R. M., Ziehn, T., Matear, R. J., Lenton, A., Chamberlain, M. A., Stevens, L. E., Wang, Y. P.,

- 318 Srbinovsky, J., Bi, D., Yan, H., & Vohralik, P. F. (2017). The carbon cycle in the Australian
- 319 Community Climate and Earth System Simulator (ACCESS-ESM1) Part 1: Model
- 320 description and pre-industrial simulation. *Geoscientific Model Development*, 10(7), 2567–2590.
- 321 https://doi.org/10.5194/GMD-10-2567-2017
- 322 Lawrence, D. M., Fisher, R. A., Koven, C. D., Oleson, K. W., Swenson, S. C., Bonan, G., Collier,
- 323 N., Ghimire, B., van Kampenhout, L., Kennedy, D., Kluzek, E., Lawrence, P. J., Li, F., Li,
- H., Lombardozzi, D., Riley, W. J., Sacks, W. J., Shi, M., Vertenstein, M., ... Zeng, X. (2019).
- 325 The Community Land Model Version 5: Description of New Features, Benchmarking, and
- 326 Impact of Forcing Uncertainty. Journal of Advances in Modeling Earth Systems, 11(12), 4245-
- 327 4287. https://doi.org/10.1029/2018MS001583

- 328 MacDougall, A. H., Frölicher, T. L., Jones, C. D., Rogelj, J., DamonMatthews, H., Zickfeld, K.,
- 329 Arora, V. K., Barrett, N. J., Brovkin, V., Burger, F. A., Eby, M., Eliseev, A. V., Hajima, T.,
- 330 Holden, P. B., Jeltsch-Thömmes, A., Koven, C., Mengis, N., Menviel, L., Michou, M., ...
- 331 Ziehn, T. (2020). Is there warming in the pipeline? A multi-model analysis of the Zero
- 332 Emissions Commitment from CO2. *Biogeosciences*, *17*(11), 2987–3016.
- 333 https://doi.org/10.5194/bg-17-2987-2020
- 334 MacDougall, A. H., Mallett, J., Hohn, D., & Mengis, N. (2022). Substantial regional climate
- 335 change expected following cessation of CO 2 emissions . *Environmental Research Letters*,
- 336 17(11), 114046. https://doi.org/10.1088/1748-9326/ac9f59
- 337 Matthews, D. H., Tokarska, K. B., Rogelj, J., Smith, C. J., MacDougall, A. H., Haustein, K.,
- 338 Mengis, N., Sippel, S., Forster, P. M., & Knutti, R. (2021). An integrated approach to
- 339 quantifying uncertainties in the RCB. *Communications Earth and Environment*, 2(1).
- 340 https://doi.org/10.1038/s43247-020-00064-9
- 341 Matthews, H. D., Tokarska, K. B., Nicholls, Z. R. J., Rogelj, J., Canadell, J. G., Friedlingstein, P.,
- 342 Frölicher, T. L., Forster, P. M., Gillett, N. P., Ilyina, T., Jackson, R. B., Jones, C. D., Koven,
- 343 C., Knutti, R., MacDougall, A. H., Meinshausen, M., Mengis, N., Séférian, R., & Zickfeld,
- 344 K. (2020). Opportunities and challenges in using RCBs to guide climate policy. *Nature*

345 Geoscience, 13(12), 769–779. https://doi.org/10.1038/s41561-020-00663-3

- 346 Matthews, H. D., & Zickfeld, K. (2012). Climate response to zeroed emissions of greenhouse
- 347 gases and aerosols. In *Nature Climate Change* (Vol. 2, Issue 5, pp. 338–341).
- 348 https://doi.org/10.1038/nclimate1424
- 349 Mauritsen, T., Bader, J., Becker, T., Behrens, J., Bittner, M., Brokopf, R., Brovkin, V., Claussen,
- 350 M., Crueger, T., Esch, M., Fast, I., Fiedler, S., Fläschner, D., Gayler, V., Giorgetta, M., Goll,
- 351 D. S., Haak, H., Hagemann, S., Hedemann, C., ... Roeckner, E. (2019). Developments in

- 352 the MPI-M Earth System Model version 1.2 (MPI-ESM1.2) and Its Response to Increasing
- 353 CO2. Journal of Advances in Modeling Earth Systems, 11(4), 998–1038.
- 354 https://doi.org/10.1029/2018MS001400
- 355 Milinski, S., Maher, N., & Olonscheck, D. (2020). How large does a large ensemble need to be?
 356 *Earth System Dynamics*, *11*(4), 885–901. https://doi.org/10.5194/esd-11-885-2020
- Mitchell, D., James, R., Forster, P. M., Betts, R. A., Shiogama, H., & Allen, M. (2016). Realizing
 the impacts of a 1.5 °C warmer world. *Nature Climate Change 2016 6:8*, 6(8), 735–737.
- 359 https://doi.org/10.1038/nclimate3055
- 360 Sellar, A. A., Jones, C. G., Mulcahy, J. P., Tang, Y., Yool, A., Wiltshire, A., O'Connor, F. M.,
- 361 Stringer, M., Hill, R., Palmieri, J., Woodward, S., de Mora, L., Kuhlbrodt, T., Rumbold, S.
- 362 T., Kelley, D. I., Ellis, R., Johnson, C. E., Walton, J., Abraham, N. L., ... Zerroukat, M.
- 363 (2019). UKESM1: Description and Evaluation of the U.K. Earth System Model. *Journal of*
- 364 Advances in Modeling Earth Systems, 11(12), 4513–4558.
- 365 https://doi.org/10.1029/2019MS001739
- 366 Seshadri, A. K. (2017). Origin of path independence between cumulative CO₂
- 367 emissions and global warming. *CIDy*, *49*(9–10), 3383–3401.
- 368 https://doi.org/10.1007/S00382-016-3519-3
- 369 Sherwood, S. C., Sen Gupta, A., & Schwartz, S. E. (2022). Probability of committed warming
- 370 exceeding 1.5 °c and 2.0 °c Paris targets. *Environmental Research Letters*, 17(6).
- 371 https://doi.org/10.1088/1748-9326/ac6ff6
- 372 Swart, N. C., Cole, J. N. S., Kharin, V. V., Lazare, M., Scinocca, J. F., Gillett, N. P., Anstey, J.,
- 373 Arora, V., Christian, J. R., Hanna, S., Jiao, Y., Lee, W. G., Majaess, F., Saenko, O. A., Seiler,
- 374 C., Seinen, C., Shao, A., Sigmond, M., Solheim, L., ... Winter, B. (2019). The Canadian

- 375 Earth System Model version 5 (CanESM5.0.3). Geoscientific Model Development, 12(11), 4823–
- 376 4873. https://doi.org/10.5194/GMD-12-4823-2019
- 377 Tjiputra, J. F., Schwinger, J., Bentsen, M., Morée, A. L., Gao, S., Bethke, I., Heinze, C., Goris,
- 378 N., Gupta, A., He, Y.-C., Olivié, D., Seland, Ø., & Schulz, M. (2020). Ocean
- 379 biogeochemistry in the Norwegian Earth System Model version 2 (NorESM2). *Geoscientific*
- 380 *Model Development*, *13*(5), 2393–2431. https://doi.org/10.5194/gmd-13-2393-2020
- 381 UNFCCC. (2015). ADOPTION OF THE PARIS AGREEMENT Paris Agreement text English.
- 382 Ziehn, T., Chamberlain, M. A., Law, R. M., Lenton, A., Bodman, R. W., Dix, M., Stevens, L.,
- 383 Wang, Y. P., & Srbinovsky, J. (2020). The Australian Earth System Model: ACCESS-
- 384 ESM1.5. Journal of Southern Hemisphere Earth Systems Science, 70(1), 193–214.
- 385 https://doi.org/10.1071/ES19035
- 386
- 387

388